Author's response to reviews

Title: Exploring the Determinants of Suicidal Behavior: Conventional and Emergent Risk (DISCOVER): A feasibility study

Authors:

Zainab Samaan (samaanz@mcmaster.ca)
Monica Bawor (baworm@mcmaster.ca)
Brittany B Dennis (dennisbb@mcmaster.ca)
Wala El-Sheikh (wala.elsheikh@phri.ca)
Jane DeJesus (jane.dejesus@phri.ca)
Sumathy Rangarajan (sumathy.rangarajan@phri.ca)
Judith Vair (jvair@stjosham.on.ca)
Heather Sholer (hsholer@stjosham.on.ca)
Nicole Hutchinson (nhutchin@stjosham.on.ca)
Elizabeth Iordan (eiordan@stjosham.on.ca)
Pam Mackie (Pam.Mackie@phri.ca)
Shofiqul Islam (Shofiqul.Islam@phri.ca)
Mahshid Deghan (mahshid.deghhan@phri.ca)
Jennifer Brasch (jbrasch@stjosham.on.ca)
Lehana Thabane (thabanl@mcmaster.ca)

Version: 3
Date: 27 March 2015

Author's response to reviews: see over
Dear Dr. Lancaster,

March 27th, 2015

We would like to thank the reviewers for taking the time to carefully review our manuscript entitled *Exploring the Determinants of Suicide: Conventional and Emergent Risk (DISCOVER): A feasibility study* (MS: 1282488266146788) and the helpful suggestions the reviewers made. We feel the manuscript in its revised form will be of high interest to the audience of Pilot and Feasibility Studies. We summarize our response to the reviewers below.

**Reviewer Comment 1:** I am very happy with the authors' responses to my review comments, and only have one small request for clarification. Discretionary Revision I only have one comment- page 6, 2nd paragraph. The authors refer to SA - this is not in their glossary, but I think this is either shorthand for suicide attempt, or they mean SB, suicidal behaviour.

**Author Response to Reviewer Comment 1:** We are glad the revised version has satisfied the reviewer. The reviewer is correct; SA is an abbreviation for suicide attempt. We have revised page 6 paragraph 2 from SA to suicide attempt. We thank the reviewer for pointing out this mistake.

We sincerely thank the reviewers for providing excellent and constructive feedback. We hope we have addressed all of their concerns.

Sincerely,

Zainab Samaan, MBChB, MSc, PhD, MRCPsych
Associate Professor of Psychiatry and Behavioural Neurosciences
McMaster University
1200 Main Street West, MDCL Room 3200, Hamilton, Ontario, Canada L8N 3Z5
Tel: 905-527-4322 Ext. 26802
Fax: 905-528-2814